Author’s response to reviews

Title: The Short-Term Economic Burden of Gestational Diabetes Mellitus in Italy.

Authors:
Michela Meregaglia (michela.meregaglia@unibocconi.it)
Livia Dainelli (Livia.Dainelli@rdls.nestle.com)
Helen Banks (helen.banks@unibocconi.it)
Chiara Benedetto (chiara.benedetto@unito.it)
Patrick Detzel (Patrick.Detzel@rdls.nestle.com)
Giovanni Fattore (giovanni.fattore@unibocconi.it)

Version: 2 Date: 08 Feb 2018

Author’s response to reviews:

1) Please include a section titled "Abbreviations" after the Conclusions section, and list here all the abbreviations used in the manuscript.

A list of all abbreviation has been included after the Conclusions.

2) Figures and legends:

• Please provide figure titles/legends under a separate heading of 'Figure Legends' after the References. If Figure titles/legends are within the main text of the manuscript, please move them.

A figure Legends heading has been added after the References.

• Figure files should contain only the image/graphic, as well as any associated keys/annotations. If titles/legends are present within the figure files, please remove them.

Title and legends have been removed from the figure files.

• Note that each figure should be uploaded as separate file in the submission system, and fit on a single A4 page in portrait format.

All figures are on single A4 pages in portrait format and uploaded separately.
3) At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colors. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

A clean version of the manuscript has been uploaded.